AUTHOR’S RESPONSE TO REVIEWS

Title: COMMUNICATION SKILLS INTERVENTION: PROMOTING EFFECTIVE COMMUNICATION BETWEEN NURSES AND MECHANICALLY VENTILATED PATIENTS

Authors:

KS Dithole (oyedeea@unisa.ac.za)
Gloria Thupayagale-Tshweneagae (tshweg@unisa.ac.za)
Oluwaseyi Akpor (akporoa@abuad.edu.ng)
Mary Moleki (Molekmm@unisa.ac.za)

Version: 3 Date: 07 Nov 2017

Authors’ response to Editorial review:

1. In your “consent for publication” section of your declarations, please confirm whether your participants gave consent for direct quotes from their interviews to be used in this manuscript.

Done
2. We would also like to ask for you to provide more justification for the contributions of OAA, as currently they do not automatically qualify for authorship. Contribution to manuscript writing alone, does not usually justify authorship.

From inception (10/05/2017) it was indicated under Authors’ contribution that OAA contributed to data analysis, drafting of the manuscript and critical revisions for intellectual content.

OAA is the corresponding author.

Author contributions:

Conception and design: KSD, GTT, MMM

Data collection: KSD, GTT, MMM

Data analysis: KSD, GTT, OAA

Drafting of the manuscript: OAA, GTT

Critical revisions for important intellectual content: KSD, GTT, OAA and MMM

3. Please remove the response to the reviewers’ comments from the file inventory as it is no longer needed at this stage of the editorial process.

Done

4. Please include your Title Page within your main manuscript, as opposed to as a separate file

Title page included in the manuscript

5. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

Done

6. We note that the current submission contains some textual overlap with other previously published works, in particular:


This overlap is spotted throughout your manuscript.
While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work.

Please re-phrase these sections to minimise overlap.

Introduction, the discussion and part of the quotations rephrased.

7. Please move your declarations to above your references.

The sections you need to include are:

Abbreviations

Ethics approval and consent to participate

Consent for publication

Availability of data and material

Competing interests

Funding

Authors' contributions

Acknowledgements

Currently, you have two competing interests section, so please remove one.

You are also missing the "ethical approval and consent to participate" section. Please include all ethical approval information in this section, along with all participant consent information.

All the above comments have been addressed as requested.

8. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.